Corresponding Author: Anouk J. M. Rombouts, MD, Department of Surgery, Radboud University Medical Centre, HP 618, PO Box 9101, 6500 HB Nijmegen, the Netherlands (anouk.rombouts@radboudumc.nl).

Published Online: January 24, 2019. doi:10.1001/jamaoncol.2018.6643

Author Contributions: Dr Rombouts had full access to all the data in the study and takes responsibility for the integrity of the data and the accuracy of the data analysis.

Concept and design: Rombouts, Hugen, Nagtegaal, de Wilt.

Acquisition, analysis, or interpretation of data: All authors.

Drafting of the manuscript: Rombouts, Huising, Hugen, Siesling, Nagtegaal, de Wilt.

Critical revision of the manuscript for important intellectual content: Rombouts, Hugen, Siesling, Poortmans, Nagtegaal, de Wilt.

Statistical analysis: Rombouts, Huising, Siesling.

Obtained funding: Nagtegaal.

Administrative, technical, or material support: Siesling, Poortmans, de Wilt. Supervision: Hugen, Poortmans, Nagtegaal, de Wilt.

Conflict of Interest Disclosures: None reported.

Meeting Presentation: This study was presented at the 71st Annual Meeting of the Society of Surgical Oncology; March 22, 2018; Chicago, Illinois.

Additional Contributions: S. J. Hogewoning, MD, PhD, Department of Research, Netherlands Comprehensive Cancer Organization, Utrecht, provided support with data handling and analysis, for which he was not compensated.

1. Depla AL, Scharloo-Karels CH, de Jong MA, et al. Treatment and prognostic factors of radiation-associated angiosarcoma (RAAS) after primary breast cancer: a systematic review. *Eur J Cancer*. 2014;50(10):1779-1788. doi:10.1016/j. ejca.2014.03.002

2. Torres KE, Ravi V, Kin K, et al. Long-term outcomes in patients with radiation-associated angiosarcomas of the breast following surgery and radiotherapy for breast cancer. *Ann Surg Oncol.* 2013;20(4):1267-1274. doi:10. 1245/s10434-012-2755-y

3. Fine JP, Gray RJ. A proportional hazards model for the subdistribution of a competing risk. JAm Stat Assoc. 1999;94(446):496-509. doi:10.1080/01621459.1999.10474144

4. Yap J, Chuba PJ, Thomas R, et al. Sarcoma as a second malignancy after treatment for breast cancer. *Int J Radiat Oncol Biol Phys.* 2002;52(5):1231-1237. doi:10.1016/S0360-3016(01)02799-7

5. Huang J, Mackillop WJ. Increased risk of soft tissue sarcoma after radiotherapy in women with breast carcinoma. *Cancer*. 2001;92(1):172-180. doi: 10.1002/1097-0142(20010701)92:1<172::AID-CNCR1306>3.0.CO;2-K

6. Linthorst M, van Geel AN, Baartman EA, et al. Effect of a combined surgery, re-irradiation and hyperthermia therapy on local control rate in radio-induced angiosarcoma of the chest wall. *Strahlenther Onkol.* 2013;189(5):387-393. doi: 10.1007/s00066-013-0316-3

COMMENT & RESPONSE

Inappropriate Grading of Adverse Events in Cancer Clinical Trials

To the Editor Consideration of toxic effects is a critical part of cancer treatment. Therefore, the grading and reporting of all grades of adverse events, especially high-grade adverse events, must be standardized to allow for consistency and comparison across trials. The Common Terminology Criteria for Adverse Events (CTCAE) has been widely used for reporting of adverse events in oncology journals and at meetings.¹

The recent study by Cristina et al² demonstrated that women with colorectal cancer receiving chemotherapy had a clinically relevant greater risk of nonhematologic and objectively measureable hematologic adverse events compared with men. The authors reported that, in 2974 patients receiving chemotherapy, 18 women (1.4%) vs 6 men (0.4%) had grade 3 or 4 alopecia (P < .003).² However, in the CTCAE, which was also used for grading adverse events in this important trial, the maximum grade for alopecia is grade 2. Thus, this is one example of the misuse of CTCAE and should be corrected immediately.¹

In our previous study, we found that a substantial proportion of contemporary cancer clinical trials use inappropriate grading and reporting of adverse events that does not conform to CTCAE.^{1,3} This situation is more urgent for trials evaluating toxic effects data, such as the study by Cristina et al.

Sheng Zhang, MD Hongxi Xue, MD

Author Affiliations: Fudan University Shanghai Cancer Center, Shanghai, China (Zhang); Rizhao City Hospital of Traditional Chinese Medicine, Rizhao, China (Xue).

Corresponding Author: Sheng Zhang, MD, Fudan University Shanghai Cancer Center, 270 Dongan Rd, 200032 Shanghai, China (wozhangsheng@hotmail. com).

Published Online: December 13, 2018. doi:10.1001/jamaoncol.2018.5849

Conflict of Interest Disclosures: None reported.

1. Zhang S, Liang F, Tannock I. Use and misuse of common terminology criteria for adverse events in cancer clinical trials. *BMC Cancer*. 2016;16:392. doi:10. 1186/s12885-016-2408-9

2. Cristina V, Mahachie J, Mauer M, et al. Association of patient sex with chemotherapy-related toxic effects: a retrospective analysis of the PETACC-3 trial conducted by the EORTC gastrointestinal group. *JAMA Oncol.* 2018;4(7): 1003-1006. doi:10.1001/jamaoncol.2018.1080

3. Zhang S. Problematic analysis and inadequate toxicity data in phase III apatinib trial in gastric cancer. *J Clin Oncol*. 2016;34(31):3821. doi:10.1200/JCO. 2016.67.3889

In Reply We thank Dr Zhang and colleagues for their comment regarding the grading of adverse events in the phase 3 PETACC-3 trial, which compared biweekly infusional fluorouracil/leucovorin alone or in combination with irinotecan as adjuvant treatment of stage 3 colon cancer.¹ We agree with Dr Zhang that the quality of adverse event grading is a critical issue in clinical trials, especially when evaluating factors influencing toxic effects. However, in contrast to what Zhang and colleagues assume, the investigators of the PETACC-3 trial¹ graded alopecia correctly as mentioned in our article²; adverse events were graded according to the National Cancer Institute of Canada Common Toxicity Grading expanded common toxicity criteria (version revised in May 1991, which was detailed in the PETACC-3 protocol but is no longer available for reference). In this version, alopecia is graded from grade 1 (mild hair loss) to grade 3 (total body hair loss). Therefore, no correction is needed; in 2974 patients receiving chemotherapy, 1.4% of women vs 0.4% of men presented with an alopecia grade greater than 2.² It is in the Common Toxicity Criteria version 2.0 and later that alopecia was graded as either grade 1 (mild hair loss) or grade 2 (pronounced hair loss).³ Moreover, we would like to mention that, regardless of the number of categories in which alopecia is graded, a bias in the comparative assessment of the severity of alopecia between men and women is likely to be introduced by differences in baseline condition, with men more often having alopecia at baseline. Furthermore, sex differences in the subjective perception of the

jamaoncology.com

severity of alopecia might also contribute to a bias because women might pay more attention to their hair loss. However, the significant differences in neutropenia described in our article clearly indicate that significant sex differences are present in chemotherapy toxic effects, including objectively measurable toxic effects.

Valerie Cristina, MD Murielle Mauer, PhD Anna Dorothea Wagner, MD

Author Affiliations: Department of Oncology, Lausanne University Hospital, Lausanne, Switzerland (Cristina, Wagner); Statistics Department, EORTC Headquarters, Brusells, Belgium (Mauer).

Corresponding Author: Anna Dorothea Wagner, MD, Department of Oncology, University Hospital of Lausanne, Rue du Bugnon 46, 1011 Lausanne, Switzerland (Dorothea.wagner@chuv.ch).

Published Online: December 13, 2018. doi:10.1001/jamaoncol.2018.5871

Conflict of Interest Disclosures: Dr Wagner reports personal fees from Merck, Merck Sharp & Dohme, Lilly, Celgene, Bristol-Myers Squibb, AstraZeneca, Ipsen, and Sanofi as well as grants from Roche outside the submitted work. No other disclosures were reported.

1. Van Cutsem E, Labianca R, Bodoky G, et al. Randomized phase III trial comparing biweekly infusional fluorouracil/leucovorin alone or with irinotecan in the adjuvant treatment of stage III colon cancer: PETACC-3. *J Clin Oncol.* 2009;27(19):3117-3125. doi:10.1200/JCO.2008.21.6663

2. Cristina V, Mahachie J, Mauer M, et al. Association of patient sex with chemotherapy-related toxic effects: a retrospective analysis of the PETACC-3 trial conducted by the EORTC Gastrointestinal Group. *JAMA Oncol.* 2018;4(7): 1003-1006. doi:10.1001/jamaoncol.2018.1080

3. National Cancer Institute. CTEP cancer therapy evaluation program. Common terminology for adverse events (CTCAE). https://ctep.cancer.gov/ protocolDevelopment/electronic_applications/ctc.htm. Accessed September 26, 2018.

Assessing the Prognostic Value of the Automated Bone Scan Index for Prostate Cancer

To the Editor Armstrong and colleagues¹ conducted an important study to assess the use of the automated Bone Scan Index (aBSI) as a tool in predicting, among other outcomes, overall survival (OS) in patients with metastatic prostate cancer. As a continuous prediction score, the aBSI's hazard ratio (HR) for OS was statistically significant, with a lower aBSI associated with better survival. However, evaluations based on HRs are difficult to interpret clinically.²⁻⁴ Moreover, the concordance index for the discriminative ability of the aBSI was only 0.63, which suggests that the continuous aBSI score may not be an effective prediction tool at the individual patient level. In practice, one may use the aBSI to stratify patients into several ordered categories. When stratifying by quartiles of the aBSI score, Armstrong and colleagues¹ reported observed median OS times (lowest to highest aBSI quartiles) of 34.7, 27.3, 21.7, and 13.3 months, respectively. These values appear to demonstrate the clinically interpretable discriminative ability of the stratified aBSI. Unfortunately, statistical inference for median times across strata was not provided. To further investigate, we generated 95% CIs of median OS times for patients in quartile 1 to quartile 4 using reconstructed data from the Kaplan-Meier curves in Figure 2A.¹ Some CIs (quartile 2 and quartile 3) overlapped. That is, the true median OS times of patients in quartile 2 and quartile 3 may be identical, suggesting

that the stratified aBSI may lack discriminative capability. The wide, overlapping CIs likely resulted from unstable median time estimates.

An alternative assessment method for the discriminative ability of aBSI may be based on restricted mean survival times (RMST).²⁻⁴ The RMSTs for 36-month follow-up are 26.4, 24.7, 21.9, and 16.0 months. The difference between quartile 1 and quartile 4 is 10.4 months (95% CI, 8.0-12.8 months; P < .001); the difference between quartile 2 and quartile 4 is 8.8 months (95% CI, 6.3-11.2 months; P < .001); and the difference between quartile 3 and quartile 4 is 6.0 months (95% CI, 3.5-8.4 months; P < .001). Over a 36-month follow-up period, a patient with an aBSI in quartile 1 would live for 26.4 months on average. Moreover, a patient with an aBSI in quartile 1 would, on average, live for 10.4 more months than a patient with an aBSI in quartile 4. The RMST provides a clinically and statistically informative assessment of the stratified aBSI.

In predictive medicine, one usually generates several categories/strata from a continuous score system to construct a stratification procedure. A desirable stratification scheme would possess small intrastratum variation and clinically meaningful discriminatory capability across strata. The usual practice of using score quartiles to group patients may be suboptimal. This issue has been discussed and alternatives have been proposed in a recently published study.⁵

Ryan Sun, PhD Huili Zhu, MD Lee-Jen Wei, PhD

Author Affiliations: Department of Biostatistics, Harvard T. H. Chan School of Public Health, Boston, Massachusetts (Sun, Wei); Department of Medicine, Icahn School of Medicine at Mount Sinai, New York, New York (Zhu).

Corresponding Author: Lee-Jen Wei, PhD, Department of Biostatistics, Harvard T. H. Chan School of Public Health, 655 Huntington Ave, Boston, MA 02115 (wei@hsph.harvard.edu).

Published Online: December 13, 2018. doi:10.1001/jamaoncol.2018.5857

Conflict of Interest Disclosures: None reported.

1. Armstrong AJ, Anand A, Edenbrandt L, et al. Phase 3 assessment of the automated bone scan index as a prognostic imaging biomarker of overall survival in men with metastatic castration-resistant prostate cancer: a secondary analysis of a randomized clinical trial. *JAMA Oncol.* 2018;4(7):944-951. doi:10.1001/jamaoncol.2018.1093

2. Uno H, Claggett B, Tian L, et al. Moving beyond the hazard ratio in quantifying the between-group difference in survival analysis. *J Clin Oncol*. 2014;32(22):2380-2385. doi:10.1200/JCO.2014.55.2208

3. Pak K, Uno H, Kim DH, et al. Interpretability of cancer clinical trial results using restricted mean survival time as an alternative to the hazard ratio. *JAMA Oncol.* 2017;3(12):1692-1696. doi:10.1001/jamaoncol.2017.2797

4. A'Hern RP. Restricted mean survival time: an obligatory end point for time-to-event analysis in cancer trials? *J Clin Oncol*. 2016;34(28):3474-3476. doi:10.1200/JCO.2016.67.8045

5. Yong FH, Tian L, Yu S, Cai T, Wei LJ. Optimal stratification in outcome prediction using baseline information. *Biometrika*. 2016;103(4):817-828. doi:10. 1093/biomet/asw049

In Reply We thank Sun and colleagues for their interest and comments regarding our work.¹ We agree that restricted mean survival time (RMST) analysis is a recent and valuable